Spatial and temporal variability of Atlantic Water in the Arctic from observations

Alice Elizabeth Richards^{1,1}, Helen Louise Johnson^{1,1}, and Camille Lique^{2,2}

¹University of Oxford ²Laboratoire d'Océanographie Physique et Spatiale

November 30, 2022

Abstract

Atlantic Water (AW) is the largest reservoir of heat in the Arctic Ocean, isolated from the surface and sea-ice by a strong halocline. In recent years AW shoaling and warming are thought to have had an increased influence on sea-ice in the Eurasian Basin. In this study we analyse 59000 profiles from across the Arctic from the 1970s to 2018 to obtain an observationally-based pan-Arctic picture of the AW layer, and to quantify temporal and spatial changes. The potential temperature maximum of the AW (the AW core) is found to be an easily detectable, and generally effective metric for assessments of AW properties, although temporal trends in AW core properties do not always reflect those of the entire AW layer. The AW core cools and freshens along the AW advection pathway as the AW loses heat and salt through vertical mixing at its upper bound, as well as via likely interaction with cascading shelf flows. In contrast to the Eurasian Basin, where the AW warms (by approximately 0.7°C between 2002 and 2018) in a pulse-like fashion and has an increased influence on upper ocean heat content, AW in the Canadian Basin cools (by approximately 0.1°C between 2008 and 2018) and becomes more isolated from the surface due to the intensification of the Beaufort Gyre. These opposing AW trends in the Eurasian and Canadian Basins of the Arctic over the last 40 years suggest that AW in these two regions may evolve differently over the coming decades.

Spatial and temporal variability of Atlantic Water in the Arctic from forty years of observations

Alice E. Richards¹, Helen L. Johnson¹, Camille Lique²

¹Department of Earth Sciences, University of Oxford, Oxford, United Kingdom ²Université de Bretagne Occidentale, CNRS, IRD, Ifremer, Laboratoire d'Océanographie Physique et Spatiale, IUEM, Brest, France

Key Points:

1

2

3

4 5

6

7

| 8 | • Atlantic Water is evolving in opposing ways in eastern and western sectors |
|----|---|
| 9 | • Data suggest Atlantic Water cools during transit via vertical mixing at its upper |
| 10 | bound and through interaction with cool dense shelf waters |
| 11 | • Atlantic Water core temperature is generally effective in assessing Atlantic Wa- |
| 12 | ter heat content but does not always capture temporal trends |

Corresponding author: Alice Richards, alice.richards@stx.ox.ac.uk

13 Abstract

Atlantic Water (AW) is the largest reservoir of heat in the Arctic Ocean, isolated from 14 the surface and sea-ice by a strong halocline. In recent years AW shoaling and warm-15 ing are thought to have had an increased influence on sea-ice in the Eurasian Basin. In 16 this study we analyse 59000 profiles from across the Arctic from the 1970s to 2018 to ob-17 tain an observationally-based pan-Arctic picture of the AW layer, and to quantify tem-18 poral and spatial changes. The potential temperature maximum of the AW (the AW core) 19 is found to be an easily detectable, and generally effective metric for assessments of AW 20 properties, although temporal trends in AW core properties do not always reflect those 21 of the entire AW layer. The AW core cools and freshens along the AW advection path-22 way as the AW loses heat and salt through vertical mixing at its upper bound, as well 23 as via likely interaction with cascading shelf flows. In contrast to the Eurasian Basin, 24 where the AW warms (by approximately 0.7 °C between 2002 and 2018) in a pulse-like 25 fashion and has an increased influence on upper ocean heat content, AW in the Cana-26 dian Basin cools (by approximately 0.1 °C between 2008 and 2018) and becomes more 27 isolated from the surface due to the intensification of the Beaufort Gyre. These oppos-28 ing AW trends in the Eurasian and Canadian Basins of the Arctic over the last 40 years 29 suggest that AW in these two regions may evolve differently over the coming decades. 30

³¹ Plain Language Summary

A few hundred meters beneath the surface of the Arctic Ocean lies a warm, salty 32 layer of Atlantic origin, called Atlantic Water (AW), which is isolated from sea-ice and 33 the ocean surface by a vertical salinity gradient that acts as a barrier between the AW 34 and the surface. In recent years, weakening of this barrier and warming of AW in the 35 eastern Arctic have contributed to unprecedented sea-ice loss. This study analyses 59000 36 vertical temperature and salinity profiles from the Arctic Ocean from the 1970s to 2018 37 to obtain a broad picture of the AW and its variability. The AW temperature maximum 38 is found to be an easily observable, generally effective way to assess how much heat is 39 stored in the AW layer. Over the period studied, the AW in the eastern Arctic warmed 40 and had an increasing influence on the amount of heat in the surface layer, whereas AW 41 heat became increasingly isolated from the surface in the west due to changes in local 42 winds. The emergence of a characteristically different eastern and western Arctic Ocean 43 in the future could have important consequences, both in terms of Arctic sea-ice loss and 44 global ocean circulation. 45

46 1 Introduction

Beneath the cool, fresh surface layer of the Arctic Ocean lies a warm, saline inter-47 mediate layer of Atlantic origin. This Atlantic Water (AW) flows in through the Fram 48 Strait (as the Fram Strait Branch) and the Barents Sea (as the deeper, cooler Barents 49 Sea Branch) and travels cyclonically around the Arctic as a topographically steered bound-50 ary current following the continental slope, with part of the current recirculating along 51 the Lomonosov and Alpha-Mendeleev Ridges into the interior and back towards the Fram 52 Strait (Aksenov et al., 2011; Woodgate et al., 2001). It eventually exits the Arctic into 53 the North Atlantic via the Canadian Arctic Archipelago (CAA) and the Fram Strait, fresher and cooler than it came in, having taken about 20-30 years to complete its journey (Lique 55 et al., 2010; M. J. Karcher & Oberhuber, 2002; Rudels, 2015; Wefing et al., 2020). Heat 56 is transferred from this AW boundary current to the interior via intrusions and eddies 57 (McLaughlin et al., 2009; Kuzmina et al., 2011). 58

The AW layer is the most significant reservoir of heat in the Arctic Ocean (Carmack et al., 2015), therefore changes in its temperature could have a significant impact on the Arctic region. The AW layer currently contains enough heat to melt all Arctic sea-ice within just a few years if this heat were brought to the surface in that time (Turner, 2010),

although across most of the Arctic the AW is isolated from the sea-ice and surface mixed 63 layer by a strong halocline. Observations suggest that AW temperature variations are 64 dominated by low-frequency oscillations with a period of 50-80 years, linked to changes 65 in the Nordic Seas which are advected through the Fram Strait (Polyakov et al., 2004). 66 Superimposed on these low-frequency oscillations are inter-annual pulse-like tempera-67 ture variations which enter through the Fram Strait or St Anna Trough and are advected 68 with the boundary current (M. J. Karcher et al., 2003; Schauer et al., 2002; Dmitrenko 69 et al., 2008; Polyakov et al., 2004; McLaughlin et al., 2009). There was also a net warm-70 ing trend in AW temperature over the twentieth century (Polyakov et al., 2004, 2012), 71 and AW in the Fram Strait is now unprecedentedly warm compared to the last two mil-72 lennia, with a rapid temperature increase in the upper AW layer over the last 120 years 73 (Spielhagen et al., 2011). 74

In the eastern Eurasian Basin, recent increases in AW temperature, along with as-75 sociated shoaling of the AW and a weakening halocline, have enhanced vertical heat trans-76 fer from the AW to the surface layer and have resulted in a substantial reduction in win-77 ter sea-ice formation (Lind et al., 2018; Polyakov et al., 2010, 2017). This "Atlantifica-78 tion" of this region shows how important a role AW can play in a changing Arctic. Fur-79 thermore, Atlantification and resultant sea-ice reduction can affect the AW itself in a va-80 riety of ways. The reduction of sea-ice import to the Barents Sea can cause a local in-81 crease in AW temperature, salinity, and hence density (Barton et al., 2018) and can also 82 result in local convection which has consequences for the AW layer downstream (Lique 83 & Steele, 2012; Lique et al., 2018). In the Eurasian Basin, reduced ice cover and a re-84 sultant increase in ventilation is expected to cause local decreases in AW temperature 85 and salinity (Pérez-Hernández et al., 2019). The impact of these local AW changes on 86 the wider Arctic region is not yet fully understood, but is another important part of the 87 changing role AW can play in the future Arctic environment. 88

Downstream in the Canada Basin the impact of AW on sea-ice is currently observ-89 able at the margins of the Canada Basin, with AW upwelling here (caused by wind) linked 90 to local sea-ice reduction (Ladd et al., 2016). Changes in both sea-ice cover and the in-91 tensity of the Beaufort Gyre in the interior Canada Basin can affect the AW (Lique & 92 Johnson, 2015; Lique et al., 2015). The recent spin-up of the gyre resulted in a deepen-93 ing of the underlying AW due to Ekman pumping, and a shoaling of the AW temper-94 ature maximum at the gyre margins (Zhong & Zhao, 2014). The pathway and intensity 95 of AW in the Canada Basin are affected by the surface circulation here as well (M. Karcher 96 et al., 2012; Lique et al., 2015). 97

Changes to the AW also have consequences outside of the Arctic. It is thought that the low density of the present warm AW anomalies in the Arctic could be maintained throughout their circumnavigation of the Arctic Ocean, and hence reduce the density of outflows into the North Atlantic (M. Karcher et al., 2011). The properties of the boundary current and these deep outflows that are advected into the North Atlantic have the potential to significantly influence overturning in this region - an important component of the global climate system.

Understanding how AW heat is likely to change in the future is therefore a key part 105 of predicting what will happen to the Arctic in the years to come. There is large discrep-106 ancy and bias amongst coupled climate model representations of AW in the Arctic, with 107 the AW layer generally being too deep and thick. The AW temperature biases are pri-108 marily due to inaccurate representation of sea ice coverage and surface cooling in the Bar-109 ents Sea, formation of cold and dense water in the Barents Sea, and AW inflow temper-110 111 atures through the Fram Strait (Shu et al., 2019; Ilicak et al., 2016). It is therefore particularly important to have an observational description of AW to help evaluate these 112 models, given their use in predicting future Arctic changes. 113

Oceanic observations in the Arctic are sparse and often seasonally biased, and many 114 observational studies of the Arctic focus on specific regions or transects (e.g. Anderson 115 et al. (1994); Beszczynska-Moller et al. (2012); Li et al. (2020); Lind et al. (2018); McLaugh-116 lin et al. (2009); Polyakov, Rippeth, et al. (2020)). However, the number of Arctic Ocean 117 observations has increased in recent years. This study aims to synthesise data from var-118 ious sources across the Arctic from the 1970s to 2018 to give a pan-Arctic, up-to-date 119 description of the AW layer and its impact on the water column. Diagnostics derived from 120 these observations, such as the temperature, salinity and depth at the AW temperature 121 maximum (the AW core) and AW heat content are used to characterise the spatial and 122 temporal variability of the AW and are described in section 2. The spatial variability of 123 the AW properties is investigated in section 3, with temporal variability in both the east-124 ern and western Arctic described in section 4. Observed changes in AW and heat dis-125 tribution within the water column at moorings and at repeated CTD transects are dis-126 cussed in section 5. Section 6 explores regional correlations between AW core metrics 127 and vertically integrated AW layer properties to investigate both how representative the 128 AW core temperature is of AW heat content, and regional differences in mixing. Con-129 clusions are given in section 7. 130

¹³¹ 2 Data and Methods

Conductivity-temperature-depth (CTD) observations from across the Arctic are 132 used in this study, from four different sources: the ice-tethered profiler (ITP) program 133 (Toole et al. (2011); Krishfield et al. (2008), http://www.whoi.edu/itp) and the Beau-134 fort Gyre Exploration Project (BGEP, https://www.whoi.edu/beaufortgyre) - both 135 based at the Woods Hole Oceanographic Institution - the Nansen and Amundsen Basins 136 Observational System (NABOS, https://uaf-iarc.org/nabos), and data from the NOAA 137 World Ocean Database (WOD, https://www.ncei.noaa.gov/products/world-ocean 138 -database). The WOD collates oceanic observational data from a wide range of sources. 139 The WOD data used here are those from CTD profiles, drifting buoys, and ocean sta-140 tions. Any ITP, BGEP or NABOS data were removed from the WOD dataset before use 141 to avoid duplication. The BGEP and NABOS datasets include data from both moor-142 ings and ship surveys. 143

Throughout the paper, salinity is given in Practical Salinity Units, and potential temperature (when not available directly from the observational data product), heat content and potential density are computed using the Thermodynamic Equation of Seawater 2010 (TEOS-10) (IOC et al., 2010).

All data used in this study are processed versions of the raw data gathered in the 148 field. Details of these procedures can be found in the sources referenced above, but all 149 involved calibration, sensor-correction and the removal or flagging of obviously erroneous 150 data. In addition to this initial processing, further routines were applied to the data and 151 profiles were smoothed for much of our analysis - details of which are given below. Pro-152 files with more than 10 % of data masked or flagged as suspicious were omitted from the 153 analysis and, unless otherwise stated, monthly mean data from moorings were used so 154 as not to bias any regional analysis to the mooring locations due to the relative high sam-155 pling frequency here compared to other locations. This resulted in about 59000 profiles 156 for analysis. 157

Here we define the Atlantic Water layer as that portion of the water column that lies below the halocline and has potential temperature above 0°C. The top and bottom of the layer are the 0°C crossing points, either side of the potential temperature maximum, and the AW layer thickness is the distance between them. As density is driven by salinity in the Arctic, potential temperature is effectively a passive tracer. The potential temperature maximum, referred to here as the AW core, is commonly used to fol-



Figure 1. (a) Map of the Arctic with bathymetric contours for every 1000 m shown along with relevant geographic features. (b) Spatial distribution of all AW core data points coloured by region, with east in blue, west in red, and mooring locations marked with black squares. (c) Spatial distribution of all AW core data points coloured by the year in which the measurements were obtained. (d) Time series showing number of monthly profiles, with east in blue, west in red and all Arctic data in black (some of which lay outside of both the eastern and western regions, so are not shown on map or used in analysis). (e) Time series showing number of monthly mooring profiles in the eastern and western regions.

low the circulation and transformations of AW (M. J. Karcher et al., 2003), and is the
 focus of much of this manuscript.

The AW core in a given profile is defined as the point at which the maximum potential temperature is reached, in the portion of the profile with salinity greater than 34.7 (in order to avoid surface temperature maxima), following Lique and Steele (2012). Figure 1 shows the distribution in time and space of profiles where the AW core was identified, with mooring locations shown as black squares.

To ensure the AW core identified in each profile was not an artefact of limited sam-171 pling, profiles were required to start above 100 m and cover a depth range of at least 500 172 m before being used to identify the AW core (this also eliminated data from the surround-173 ing shelf seas, allowing the study to focus on AW within the Arctic basin only). This lat-174 ter step resulted in about 44000 AW core data points. Before identifying the core, pro-175 files were smoothed over a vertical distance of 80 m by taking the mean of the profile 176 data within 40 m of each data point. This length scale was chosen as it was the best at 177 preserving the general shape and magnitude of the temperature profile, while removing 178 the spikes due to features such as thermohaline intrusions and eddies (although please 179 note that this smoothing was not applied to the profiles in Figure 3). This is important 180 as the main aim of this study is to get a general picture of patterns and long-term trends 181 in AW core properties, and features such as intrusions can disproportionately affect the 182 depth of the AW temperature maximum in basin interiors in such a way as to detract 183

from this. Any profiles used in the analysis should be assumed to be smoothed unless stated otherwise.

The depth coverage varies between data sources - ITP profiles cover the upper 800 186 m of the water column, whereas many data from CTD stations and moorings extend down 187 to around 2000 m. This variation in depth range does not affect the analysis in this study 188 given that the AW core can be identified in both cases, and any profiles that do not sam-189 ple the whole AW layer are omitted from the AW heat content analysis in sections 5 and 190 6. Although ITPs and moorings provide year-round measurements, there remains a spring/summer 191 192 bias in data from other sources. However, this is unlikely to impact results due to the negligible seasonality of AW when compared to its overall variability in space and time 193 (Lique & Steele, 2012). 194

Heat content, HC, was computed for various portions of the vertical temperature
 profiles according to

$$HC = \int_{a}^{b} \rho_{\theta}(z) c_{p} T(z) dz \tag{1}$$

where ρ_{θ} (in kgm^{-3}) is potential density, c_p (in $Jkg^{-1} \circ C^{-1}$) is specific heat, T (in $\circ C$) 197 is potential temperature, z (in m) is depth, and a and b are the depth bounds defining 198 the layer in question. Approximately 1500 profiles sampled the entire AW layer and al-199 lowed for the computation of total AW heat content. To account for differing profile lengths 200 above the AW layer (due to variation in upper depths of profiles and the depth of the 201 AW layer itself), heat content density is used to evaluate the heat stored in this upper 202 layer in a similar way to Polyakov et al. (2011). The heat content derived for this up-203 per portion of the water column is divided by the depth range over which it is computed. 204 This quantity is proportional to the average temperature over that depth range.

²⁰⁶ 3 The Atlantic Water core across the Arctic

Investigating how the hydrographic properties of the AW core change across the Arctic can give a good picture of the behaviour of the AW layer in general. Figure 2 shows maps of the potential temperature, salinity, depth and potential density anomaly of the AW core from all observations, giving an idea of the general spatial distribution of properties at the depth of the AW temperature maximum.

Figure 2b highlights the temperature difference between AW core properties in the 212 Eurasian Basin and western Arctic. The core loses heat as it is advected around the basin 213 - its temperature in the Canada Basin is approximately 1.5°C lower than where it is first 214 subducted under the fresh surface layer at the southern boundary of the Eurasian Basin. 215 The most significant heat loss is seen here, where, much like in the Nordic Seas (Lind 216 et al., 2018), the AW loses heat to the atmosphere and through mixing with the cooler 217 surface layer. The higher AW core temperature along the Lomonosov Ridge relative to 218 the western Arctic boundary suggests that the AW that recirculates back along the ridge 219 is warmer than that which continues towards the western Arctic. 220

The salinity of the core (Figure 2c) decreases on its journey around the Arctic, par-221 ticularly in the Eurasian Basin where it mixes with fresher surface waters upon subduc-222 tion. Turbulent mixing may play an important role in AW freshening in parts of the west-223 ern Arctic - the difference in AW core salinity (and temperature) between the bound-224 ary and interior of the Canada Basin is indicative of this. Whereas the AW in the in-225 terior of the Canada Basin has travelled around the north of the Chukchi Plateau, the 226 AW at the boundary has travelled over the Chuckchi Plateau's complex bathymetry (McLaughlin 227 et al., 2009; Li et al., 2020). The relatively low temperature and salinity of this bound-228 ary AW can be explained by enhanced mixing experienced over this rough topography 229 upstream. 230



Figure 2. Maps showing (a) depth (b) potential temperature (c) salinity and (d) potential density anomaly of all AW core data points used in this study.

Despite the AW core freshening along the AW advection pathway, the density of 231 the core appears to increase from its relatively low value along the southern Eurasian 232 Basin boundary to higher values in the western Arctic and Eurasian Basin interior (Fig-233 ure 2d). There are particularly dense, cold regions along the western shelves just north 234 of the Canadian Arctic Archipelago (CAA) and Greenland. This is surprising given the 235 importance of salinity in governing density at such cold temperatures, but may be ex-236 plained by heat loss from the top of the AW layer to the fresher, cooler water above through 237 vertical mixing as AW is advected. This would deepen the AW core without the AW layer 238 as a whole getting denser, as heat lost preferentially from the upper AW would result 239 in the core being found on deeper (denser) isopycnals. This could explain the cooling, 240 freshening and increase in density of the AW core seen along the AW advection path-241 way, with the coldest and densest regions towards the end of its journey (Aksenov et al., 242 2011).243

Another process which might be contributing to the cooling and freshening of the 244 AW core during its advection is interaction between AW and cold, dense cascading flows 245 from the shelves (formed from brine rejection during sea-ice formation). These flows have 246 been modelled throughout the Arctic (Luneva et al., 2020), and have been observed to 247 interact with and modify AW in the Eurasian Basin (Ivanov & Golovin, 2007). Obser-248 vations of this interaction in the western Arctic are limited due to the sparsity of obser-249 vations in these shelf regions, but dense water cascades have been observed to interact 250 with waters as deep as the lower halocline here (Ivanov et al., 2004; Melling & Moore, 251 1995; Luneva et al., 2020), suggesting that interaction with upper AW is feasible. 252

Due to limitations in our dataset, we are unable to quantify the relative importance of these two potential mechanisms through which the properties of the AW core are modlified. Although observational evidence of interaction between shelf flows and AW is limited in the western Arctic, it is likely that both heat loss from the top of the AW layer and interaction with shelf flows contribute to the AW core becoming cooler and denser along the AW advection pathway.

The AW core depth exhibits a bimodal structure, as shown in Figure 2a, being much 259 deeper in the Canada Basin (approx. 500 m) than the Eurasian Basin (approx. 300 m) 260 due to the Ekman pumping associated with the winds which drive the Beaufort Gyre. 261 The effect of the Beaufort Gyre on the AW in the Canada Basin can also be seen in the 262 (un-smoothed) vertical temperature profiles in Figure 3, where the cool waters of the gyre 263 push down the AW layer to a much greater depth than that at which it resides in the 264 eastern Arctic. However, the important role that the halocline plays in isolating the AW 265 from the surface can be observed across the whole Arctic (see Figure 3). 266

Zig-zags and staircase features in these un-smoothed profiles also indicate the pres-267 ence of thermohaline intrusions and diffusive convection, respectively (Bebieva & Tim-268 mermans, 2017). Thermohaline intrusions form in the presence of temperature and salin-269 ity gradients along isopycnals (Ruddick, 1992), and are an important mechanism for AW 270 transport from the boundary to the interior of both the Canada and Eurasian Basins (McLaughlin 271 et al., 2009; Kuzmina et al., 2011). They are often found near the AW core depth (un-272 like staircases which tend to be found above the AW core depth where diffusive convec-273 tion is supported (Bebieva & Timmermans, 2019)). These intrusion signatures are seen 274 in Canadian Basin data in the 2000s. Intrusions are also seen in the Eurasian Basin through-275 out the time period covered in this study, providing strong evidence for their long-term 276 presence in this region, although they have been seldom documented beyond the Cana-277 dian Basin in previous studies. 278



Figure 3. Un-smoothed potential temperature (solid) and salinity (dashed) profiles (a) in Canada Basin interior, (b) on Canada Basin shelf, (c) at the Siberian end of Lomonosov Ridge, (d) in the eastern Eurasian Basin, and (e) in the western Eurasian Basin. Profiles are coloured by the year in which they were measured, with year and month given in the legend. Note the change in y-axis scale in (c).

4 Regional differences in Atlantic Water properties and their temporal variability

Although the maps in Figure 2 give an idea of the general spatial variability of AW, they do not indicate how the AW has changed over the time period studied. This section will explore the temporal variability observed in the AW layer in different regions of the Arctic.

Figure 3 shows profiles from the same locations measured in different years. The 285 uppermost plot in Figure 3 uses profiles from moorings at the Siberian end of the Lomonosov 286 Ridge, sampling the AW boundary current. The depth range sampled by the moorings 287 captures both AW branches - the Fram Strait Branch Water (FSBW) and the Barents 288 Sea Branch Water (BSBW), centered at around 200-500 m and 750-1000 m, respectively. 289 The temperature of these two branches here appears to vary independently in time. The 290 BSBW shows a general warming trend throughout the period sampled (although this 291 conclusion is tentative as we cannot rule out an influence of transient features such as 292 eddies). This warming could hint at a more systematic change in BSBW temperature, 293 which could be explained by surface air temperature increases over the Barents Sea (Skagseth 294 et al., 2020) or reductions in sea-ice import to the region (Lind et al., 2018). The FSBW 295 temperature is more variable from year to year, reflecting the variability of AW inflow 296 temperature through the Fram Strait (Ivanov et al., 2012), although local transient fea-297 tures may also play a role in these profile differences. The heat loss experienced by the 298 AW in the Barents Sea may act as a buffer for BSBW against high-frequency variation in upstream AW temperature. 300

Building on the regional differences in AW shown in Figure 2, Figures 4–7 highlight how the properties of the AW core in the eastern (blue) and western (red) Arctic evolve with time. Canada Basin mooring data is shown in black. Maps and annual normalised histograms show how the potential temperature, salinity, potential density anomaly and depth of the AW core change. The period covered by each map is indicated by grey lines enveloping the corresponding annual histograms - these periods were chosen to account for the varying amount of data available during each period. The reader is referred to Figure 1 for time series of the amount of data available from each region.

The year-to-year spatial variation in data distribution in the eastern Arctic makes 309 inferring any trends from the histograms for the eastern Arctic difficult, and no signif-310 icant trend can be found. After applying our processing scheme for core detection, only 311 three years of eastern Arctic mooring data provides information on AW core properties 312 (see Figure 1), so no meaningful trends can be identified from this fixed-point data ei-313 ther. However, the more consistent spatial distribution of data in the western Arctic and 314 long time-series from mooring data allow trends to be inferred for the Canada Basin -315 these are described in more detail in the next section of this paper. The differences in 316 the range of temperature and salinity data between the east and west highlights the trans-317 formation undergone by the AW as it travels around the basin, reinforcing what was shown 318 in Figure 2, with AW core salinity and temperature decreasing due to mixing, and AW 319 core density increasing. 320

The mooring data in Figure 4 reveal a gradual cooling in the Canada Basin after 2002, presumably after the arrival of the AW core warm temperature anomaly which entered the Canada Basin in the early 2000s (after having entered the Arctic through the Fram Strait in 1990) McLaughlin et al. (2009); Li et al. (2020). The maps in Figure 4 show the spread of this anomaly from the northern edge of the Chukchi Plateau into the interior of the Canada Basin in 2000–2004, with a more homogeneous AW core temperature field in 2005–2009.

Figures 5, 6 show an increase in AW core salinity and density in the interior of the Canada Basin throughout the mooring time period (2003–2018). This could be a down-



Figure 4. Annual normalised histograms of AW core potential temperature. Histogram data is coloured by region, with blue for eastern data, red for western data, and black for western mooring data - as shown in the map at the top right. The remaining maps on the right show the spatial distribution of the data in the histograms contained within each pair of grey lines.



Figure 5. As Figure 4 for AW core salinity



Figure 6. As Figure 4 for AW core potential density anomaly



Figure 7. As Figure 4 for AW core depth

stream impact of Eurasian Basin Atlantification. Local increases in AW salinity in the
Eurasian Basin and surrounding seas due to reduced Eurasian Basin sea-ice have been
documented (e.g. (Lind et al., 2018; Barton et al., 2018)), but the impact of these changes
on AW downstream in the western Arctic is not yet fully understood.

Alternatively, an increase in the AW core salinity and density within the Canada 334 Basin could be due to the AW core moving deeper as heat is transported away from it. 335 Unlike at the boundary, where turbulent mixing (McLaughlin et al., 2009; Li et al., 2020) 336 results in similar changes in AW core temperature and salinity, in the interior diffusion 337 governs the vertical transport of temperature and salinity. The diffusive convection regime 338 in the upper AW layer (Bebieva & Timmermans, 2019) results in temperature being dif-339 fused more effectively than salt, causing the core to cool but not freshen. The warmest 340 point is then located at a deeper, more saline part of the AW. 341

The most prominent trend in these histograms is the increasing depth of the AW 342 core in the Canada Basin mooring data (Figure 7). Zhong and Zhao (2014) showed that 343 the AW deepening caused by the spin up of the Beaufort Gyre dominates over the in-344 fluence of AW core density on depth if the gyre intensifies sufficiently, with AW position 345 in relation to the gyre centre becoming more important than its density from 2007 on-346 wards. This means that when the gyre is sufficiently intense, AW suppressed by the gyre 347 can reside deeper than other AW that is denser (e.g. that at the Canada Basin bound-348 ary which the gyre does not suppress). Figure 7 shows that the deepening of the AW core 349 coincides with the isopycnal deepening reported by Zhong and Zhao (2014), Proshutinsky 350 et al. (2019) and others. The effect of the AW core potential density increase on this core 351 deepening will be investigated in the following section of this paper, where the whole wa-352 ter column will be considered. 353

³⁵⁴ 5 AW variability at transects and moorings

Broad regional trends in AW core properties have been discussed above. To investigate these further, and put AW core property changes within the context of the wider water column, the temporal variability of data at moorings and across regularly repeated CTD transects is investigated below. This reveals more about the implications of AW changes for water column stratification and heat distribution. Trends from individual Canada Basin moorings (black data in Figures 4–7) are also discussed in more detail.

361 5.1 Eurasian Basin

The potential temperature and salinity along a NABOS CTD transect repeated from 362 2002–2018 across the AW boundary current in the eastern Eurasian Basin is shown in 363 Figure 8. The year of each transect is given in the plot, and the AW core depth is iden-364 tified with a black dot. The vertical black lines near the surface of the transects show 365 the location of the CTD profiles. Data between these profiles have been interpolated us-366 ing a Delaunay triangulation grid. The AW layer warms in general throughout the time 367 period (when comparing the start and end years). However, this warming is pulse-like 368 rather than continuous, with one warm pulse peaking in 2007-08 (likely the same warm 369 pulse of AW that entered through Fram Strait around the year 2000 (Polyakov et al., 370 2005, 2011)) and a second in the 2018 section. The AW core in 2018 is 1°C warmer than 371 that in 2002. The salinity of the AW also shows an increasing trend throughout the time 372 period covered in Figure 8, although regions and years of high salinity are not coinci-373 dent with regions or years of high temperature. 374

Figure 9 allows for a more quantitative assessment of the changes in the water column at this location. The first three panels of this figure show the evolution of AW core properties across the transect. This figure shows more clearly that the core freshens onshore in most years, suggesting that AW is mixing with fresher waters from the shelf or



Figure 8. Potential temperature and salinity along a repeated CTD transect in the eastern Eurasian Basin. The year in which each transect was taken is given at the bottom left of each plot. In all years, transects were measured in August, September or October. The origin of the x-axis of the transect is marked with a black cross on the map, so that the x-axis origin is at the most offshore station. CTD profile locations are marked on the transect plots with vertical black lines at the surface. The black dots on the transect plots denote the location of the AW core.



Figure 9. Water column properties across a repeated CTD transect in the eastern Eurasian Basin. Year markers denote the start of that year. Transect location is shown on the map, with a black cross denoting the x-axis origin of the transect plots. The x-axis origin is therefore at the most offshore station. Transect plots show (a) AW core potential temperature, (b) AW core salinity, (c) AW core depth, (d) heat content density of the water column above the AW layer, denoted as upper heat content (UHC), and (e) total heat content of the AW layer.

that the AW that reaches the shelf is that which is fresher. As above, the core temper-379 ature (and AW layer heat content shown in panel five) exhibits warm pulses which are 380 superimposed upon a general warming trend across the period. Between 200-300 km along 381 the transect, AW core temperature and salinity have linear trends of 4.13×10^{-2} °C per 382 year (resulting in an increase of 0.7 °C between 2002–2018) and 3.63×10^{-3} psu per year, 383 respectively. Nearer the boundary, greater than 350 km along the transect, these trends 384 are reduced to 1.52×10^{-2} °C per year and 7.64×10^{-4} psu per year. This reduction 385 in magnitude of AW core property trends away from the interior is likely due to enhanced 386 mixing near the boundary. Notably the heat content of the AW layer increases to three 387 times its 2002 value in 2018. The salinity and depth of these warm pulses differ, how-388 ever - the AW core during the warm pulse in 2018 is fresher and shallower than the one 389 in 2008. A weakened halocline may have allowed the warm AW to shoal higher in the 390 water column and mix with the fresher surface layer, as reported by Polyakov, Rippeth, 391 et al. (2020). The 2013 transect, although slightly cooler than those from 2008 and 2018, 392 has a comparatively salty, deep AW core. This non-coincidence of AW core salinity and 393 temperature changes suggests that even enhanced mixing due to a weaker halocline does 394 not mask the signal of these warm AW pulses. 395

The fourth panel of Figure 9 shows the "heat content density" (the heat content 396 of a portion of the water column divided by the height over which it is computed) of the 397 sampled water column above the AW layer, denoted as upper heat content (UHC) in the 398 figure. This quantity is proportional to the average temperature of the surface layer and 399 halocline. UHC increases when the AW core salinity is low and AW core depth is shal-400 low - perhaps an indication that a shallower AW layer transfers more heat to the halo-401 cline and surface layer. Despite the similarities in AW core temperature of the 2009 and 402 late-2013 transects, the fresher/shallower core in 2009 coincides with a larger surface layer 403 heat content density. This emphasises the strong link between halocline strength and the 404 depth and salinity of AW. This is also highlighted in recent work by Polyakov et al. (2018), 405 where halocline stability, quantified using density anomalies throughout the layer, is iden-406 tified as a key climate change indicator in the region. The implication that AW shoal-407 ing is more influential than AW temperature change on surface layer heat content is not 408 surprising given the low levels of vertical mixing throughout the Arctic (Fer, 2009). This 409



Figure 10. Water column properties across two repeated CTD transects in the Canada Basin. Year markers denote the start of that of year. Transect location is shown on the maps, with black crosses denoting the x-axis origin of each transect. The x-axis origin is therefore at the most offshore station. Remaining panels show (a) AW core potential temperature, (b) AW core salinity, (c) AW core depth, (d) heat content density of the water column above the AW layer, denoted as upper heat content (UHC), and (e) total heat content of the AW layer.

is also reflected in the dissimilarity between variations in AW layer heat content and UHC seen in Figure 9.

412 5.2 Canada Basin

Figure 10 shows the same analysis applied to two repeated CTD transects in the Canada Basin. This allows for detection and comparison of any signals advected downstream from the Eurasian Basin transect in Figure 9. The length of the transects also enables comparisons between AW found on the boundary and within the interior of the Canada Basin.

As in the Eurasian Basin, Figure 10 shows evidence of the pulse-like nature of AW 418 core temperature evolution, with warm AW core values in the interior in the mid-2000s 419 indicative of the warm anomaly that arrived in the Canada Basin in the early 2000s (McLaughlin 420 et al., 2009). As seen in Figure 2, AW at the Canada Basin boundary (>1000 km along 421 section A, and the furthest few data points of section B) is cooler and fresher than that 422 in the interior due to the enhanced mixing it experiences upstream over the rough bathymetry 423 of the Chukchi Plateau (McLaughlin et al., 2009; Li et al., 2020). This enhanced mix-424 ing and cooling is the likely reason why the warm temperature anomaly is not seen at 425 the boundary in Figure 10 - the temperature signal is much weaker there than it is in 426 the interior. 427

⁴²⁸ On both transects, at the boundary, AW core temperature and salinity vary sim-⁴²⁹ ilarly, presumably because both of these properties are governed by the same mixing pro-



Figure 11. Time series of monthly mean AW core (a) potential temperature, (b) salinity, (c) potential density anomaly, and (d) depth from the four Canada Basin moorings. Mooring locations (A-D) are shown on the map at the top right.

cesses upstream. This is not the case in the interior, however, where from 2012 onwards 430 both transects see an increase in AW core salinity which is not reflected in the temper-431 ature. This increase in salinity (and density) and decrease in temperature in the inte-432 rior was also seen in the Figures 4–6. T-S plots of un-smoothed profiles from the ends 433 of each annual transect (not shown here) reveal that the increase in interior salinity is 434 accompanied by a disappearance of thermohaline intrusions i.e. the commencement of 435 AW core cooling after the warm anomaly has arrived. This confirms the mechanism pro-436 posed earlier - of heat diffusing away from the AW core resulting in the core residing at 437 a more saline part of the AW layer - as the most likely explanation for these trends. The 438 idea that this AW core salinity increase is related to AW core heat loss, and is not re-439 lated to changes in the AW layer as a whole, is also supported by a lack of trend in AW 440 layer mean salinity in the Canada Basin interior over this period (not shown). 441

The slight warming in 2016 of the AW core temperature in the interior of section 442 B of Figure 10 could be evidence of the AW warm anomaly observed upstream of the 443 Chukchi Borderlands in 2010 (after having entered the Arctic Ocean through the Fram 444 Strait around 2000, Li et al. (2020)). It can also be seen at the Eurasian Basin transect 445 in Figure 9 around 2008, and until now has not been conclusively observed in the Canada 446 Basin interior. This gives AW advection timescales from the eastern Eurasian Basin to 447 the north of the Chuckchi Borderlands and around the north of the Borderlands into the 448 Canada Basin interior of order 8 years, in agreement with other observational studies 449 (Polyakov et al., 2011; Li et al., 2020). The amplitude of the warming is low compared 450 to that of the previous warm temperature anomaly, despite the second anomaly being 451 0.24°C warmer than the first in the Eurasian Basin (Polyakov et al., 2010). This could 452 be due to enhanced heat loss experienced by the AW during its advection, associated with 453 increased ventilation, enhanced mixing with cooler, fresher water above, and/or inter-454 action with shelf flows in the eastern Arctic (Ivanov & Golovin, 2007). Further obser-455 vational data would be needed to confirm the presence of this second AW warm anomaly 456 in the Canada Basin interior, however. 457

As discussed in Figures 4–7, the AW core in the Canada Basin cooled, freshened and deepened from the early 2000s onward. Trends from individual Canada Basin moorings (data in black in Figures 4–7) are more clearly shown in the time series in Figure

Table 1. Trends in monthly mean AW core potential temperature, salinity, potential densityanomaly, and depth from each of the four Canada Basin moorings. R-squared values for the fit ofeach trend are given in parentheses. Note that mooring C data only cover five years (see Figure 11).

| Mooring | θ trends / °C $year^{-1}$ | Salinity trends / psu $year^{-1}$ | σ_{θ} trends / $kgm^{-3}year^{-1}$ | Depth trends $/ m y ear^{-1}$ |
|--------------|-------------------------------------|-----------------------------------|--|-------------------------------|
| Α | $2.81 \times 10^{-4}(0.00)$ | $2.22 \times 10^{-3}(0.76)$ | $1.79 \times 10^{-3}(0.88)$ | 9.204(0.79) |
| в | $-1.81 \times 10^{-2}(0.70)$ | $6.90 \times 10^{-4}(0.26)$ | $1.70 \times 10^{-3}(0.89)$ | 12.24(0.82) |
| \mathbf{C} | $3.82 \times 10^{-2}(0.61)$ | $4.74 \times 10^{-3}(0.70)$ | $1.33 \times 10^{-3}(0.43)$ | 9.80(0.79) |
| D | $2.17 \times 10^{-4}(0.00)$ | $1.96 \times 10^{-3}(0.73)$ | $1.52 \times 10^{-3}(0.75)$ | 9.01(0.88) |

⁴⁶¹ 11 and are given explicitly along with their R-squared values in Table 1. Again, the west⁴⁶² ern AW core temperature and depth trends in Figure 11 and Table 1 oppose those ex⁴⁶³ pected from the Atlantification reported in the east.

As seen in the maps in Figure 4, the warm anomaly that entered the Canada Basin 464 in the early 2000s spread into the interior from the Chukchi Plateau before cooling (McLaughlin 465 et al., 2009). This is reflected in the consistent, significant cooling trend at mooring B 466 (Figure 11 and Table 1), and the pre-2010 warming at moorings C and D (Figure 11). 467 There is an increase in salinity at all moorings, with significant trends at moorings A, 468 C and D. As discussed above, this seems to be related to the cooling of the core after 469 the spread of the warm anomaly. The resultant increase in AW core density can be clearly 470 seen in Figure 11, with significant trends at moorings A, B and D (note that C has a shorter 471 time series). The trends in depth in both Figures 7 and 11 are also significant (Table 1), 472 and could be attributed to the increase in AW core salinity. However, comparing the pat-473 terns of change in AW core depth and salinity in Figure 10 suggests that the salinity in-474 crease is not the most important driver of this depth increase. As discussed, the spin-475 up of the Beaufort Gyre and associated enhanced downwelling influences AW depth (Lique 476 & Johnson, 2015; Lique et al., 2015; Zhong & Zhao, 2014), so the increase in core depth 477 is likely explained by gyre intensification. 478

The Hovmöller plots in Figure 12 reveal more about the mechanisms behind the 479 AW core deepening in the Canada Basin, and also help explain the increase in upper layer 480 heat content density. This figure shows Hovmöller plots of potential temperature pro-481 files from three of the BGEP moorings in the Canada Basin (mooring c having been omit-482 ted due to the shorter time series available there), with white lines marking isopycnal 483 depths. The core depth varies in concert with isopycnal depth at all moorings, again em-484 phasising that changes elsewhere in the water column, rather than those of the AW prop-485 erties themselves, are likely driving AW depth changes here. The increase in AW core 486 salinity/density does not appear to play an important role, as seen in Figure 10. With 487 the exception of mooring D (which the gyre moved away from during the start of the pe-488 riod shown (Regan et al., 2019)), the isopycnals and hence AW deepen over the time pe-489 riod covered. The temporal pattern of this deepening is in agreement with previous pa-490 pers (Proshutinsky et al., 2019; Zhang et al., 2016) and can be attributed to an overall 491 increase in Ekman pumping (and hence freshwater content) in the central Canada Basin 492 between 2003-2018 due to a combination of changes in sea-ice conditions and the strength 493 of the Beaufort Gyre (Proshutinsky et al., 2019). 494

The increase in upper layer heat content density seen in Figure 10 can also be attributed to Beaufort Gyre intensification. Pacific Water is subducted into the halocline of the interior Canada Basin via isopycnals which outcrop in the Chukchi Sea (Timmermans et al., 2014). The increase in Ekman transport has therefore resulted in thicker Pacific



Figure 12. Hovmöller plots of potential temperature from three moorings in the Canada Basin (BGEP moorings A, B and D - locations shown on map). White lines denote the isopycnals - the deepest at 27 kgm^{-3} , contour interval 1 kgm^{-3} . Grey regions cover time periods with insufficient data.

Water layers within the Canada Basin halocline (Timmermans et al., 2014), and the warm-499 ing of Pacific Summer Water (which can be seen as the warm shallow water mass that 500 appears in later years of Figure 12) between 2003-2014 has subsequently resulted in an 501 increase in halocline heat content (Timmermans et al., 2014, 2018). Changes in upper 502 layer heat content density are therefore independent of AW heat content, as seen in Fig-503 ure 10. This suggests that increases in sea-ice bottom-melt reported in the Beaufort Sea 504 (Perovich & Richter-Menge, 2015) are likely due to Pacific Summer Water (along with 505 other local features such as the near-surface temperature maxima (Jackson et al., 2012; 506 Timmermans, 2015)), not AW warming. The deepening of the AW, and its increased iso-507 lation from the surface in the Canada Basin, is in stark contrast to the concurrent At-508 lantification seen in the Eurasian Basin. 509

Although spatial and temporal patterns of AW core potential temperature and AW layer heat content in Figure 10 are similar, they do not match as closely as might be expected. We explore this further in the following section.

⁵¹³ 6 Relationships between profile metrics

Much of the analysis in this study has involved the use of AW core properties to infer AW layer properties within the Arctic. It is therefore important to investigate how representative AW core properties are of the AW layer in general. Here we compute correlations between AW core and integrated AW layer metrics, which also shed some light on how the AW layer loses heat in each region.

The maps in Figure 13 show total AW layer heat content during different time periods, chosen to give a roughly even data distribution between panels. As most observational profiles do not sample deep enough to cover the entire AW layer, there are sub-



Figure 13. Maps of total AW layer heat content from all profiles which sampled the entire AW layer (defined as the layer between the two 0 °C crossing points either side of the AW core depth) for (a) 1980-1999, (b) 2000-2004, (c) 2005-2009, (d) 2010-2014, and (e) 2015-2018.

stantially less AW heat content data (1500 data points) than AW core data. This, of it self, emphasizes the usefulness of AW core data in assessing the pathways and evolution
 of the AW layer.

In the Canada Basin, total AW layer heat content increased in the mid-2000s (Fig-525 ure 13, panels b to c) after the arrival of the AW warm anomaly (McLaughlin et al., 2009; 526 Li et al., 2020) and has since remained at that higher level of approximately 1.5×10^9 527 Jm^{-2} , with no long term trend observed. The eastern Eurasian Basin (EEB) (i.e. the 528 portion of the Eurasian Basin east of 90°E which has good data coverage for much of Fig-529 ure 13) saw an increase in AW heat content throughout the period studied, in-line with 530 the reported Atlantification of the region (Lind et al., 2018; Polyakov et al., 2010, 2017). 531 Figure 13 shows a stark difference between AW heat content in the eastern Eurasian and 532 Canada Basins, implying that the AW that bifurcates and recirculates towards the Fram 533 Strait along the Lomonosov Ridge is warmer than the AW in much of the western Arc-534 tic. The heat content maps in Figure 13 are very similar to the AW core potential tem-535 perature maps in Figure 2, further suggesting that the AW core temperature captures 536 the general pan-Arctic spatial pattern of AW heat content variability well. However, the 537 cooling of the AW core in the Canada Basin (Figures 4 and 11) does not seem represen-538 tative of the steady western AW heat content values in Figure 13, providing more ev-539 idence that the decrease in temperature (and increase in salinity) of the AW core do not 540 reflect temporal changes in the AW layer itself. 541

To be more quantitative in assessing how well the AW core temperature represents AW heat content, correlations between these two metrics were computed. Figure 14 shows scatter plots between pairs of variables computed from profiles in both the eastern (restricted to EEB only due to constraints on spatial data coverage - see Figure 13) and western Arctic. These regions are defined in Figure 1. R-squared values for each of the plots



Figure 14. Scatter plots between (a) AW core potential temperature and total AW heat content, (b) AW layer thickness and AW heat content, (c) AW core depth and AW top depth, (d) AW mean salinity and AW heat content, and (e) AW core salinity and AW core potential temperature. Blue data is from the eastern Arctic, with red data from the western Arctic (regions defined in Figure 1). R-squared values and regression lines are shown for each scatter plot.

are given, along with regression lines. The relationship between total AW heat content 547 and the potential temperature of the AW core is shown in Figure 14a. There is a strong 548 correlation between these two variables in both the EEB and western Arctic, highlight-549 ing the general effectiveness of the AW core temperature as an easily measurable met-550 ric for assessing changes in AW heat content. As noted above, however, caution must 551 be taken when using AW core properties to infer temporal trends in the AW layer, as 552 the cooling (and resultant increase in salinity) of the AW core does not necessarily cor-553 respond with a cooling (or salinification) of the AW layer more generally. 554

Figure 14b shows AW layer thickness against AW layer heat content, with a moderate correlation in the EEB and a weak correlation in the west. This is an important reminder that although the potential temperature of the AW core will give a good idea of how AW heat content may vary (see Figure 14a), AW heat located away from the AW core also affects AW heat content. This is particularly true in the EEB where BSBW heat in the deep AW (which does not vary in concert with FSBW/the AW core, as seen in Figure 3) affects AW heat content.

Figure 14c shows the relationship between AW core depth and the depth of the up-562 per boundary of the AW layer (i.e. the 0°C crossing point above the AW core depth). 563 AW temperatures in both the eastern and western Arctic appear to be affected by heat 564 loss from the top of the AW layer, causing the temperature maximum to deepen, although 565 not necessarily to the same extent as the AW top (which is defined by a fixed temper-566 ature). This lowers the correlation between these two variables. However, there is a much 567 stronger correlation in the west than in the EEB, as AW core depth and AW top depth 568 in the Canada Basin are both greatly influenced by the Beaufort Gyre which affects the 569 two metrics in the same way. The comparatively low correlation between AW core depth 570 and AW top depth in the EEB highlights the care that should be taken when using AW 571 core depth to assess AW layer shoaling here. 572

Comparing the mean salinity of the AW layer with the AW layer heat content can give an idea of the role that mixing with fresher waters plays in AW heat loss. Figure 14d shows that, while there is a relatively high correlation between these variables in the EEB, the correlation is negligible in the west. This implies that although mixing with fresher waters is important for AW heat loss in the Eurasian Basin - as would be expected given that the AW subducts beneath the cooler, fresh polar waters here, losing a lot of heat - it is not as important in the western Arctic.

In Figure 14e, AW core salinity is compared to AW core potential temperature. In the EEB this reflects what is seen in the integrated AW layer (Figure 14d). In the west however, a stronger (moderate) correlation exists between the AW core data than that between the AW layer data. This could be related to the thermohaline intrusions which spread the warm AW core anomaly into the Canada Basin during the study period (McLaughlin et al., 2009). However it also highlights once again the freshening of the AW core, relative to the AW layer, as it is advected around the Arctic.

587 7 Conclusion

This study has used all available hydrographic profiles from across the Arctic from 588 the 1970s to 2018 to build a picture of AW in the Arctic Ocean entirely from observa-589 tions, and to investigate its spatial and temporal variability. Much of the analysis has 590 focused on the properties of the AW core (the depth at which the maximum potential 591 temperature occurs). This was found to be a generally effective and easily detectable met-592 ric to assess the heat content of the AW layer. However, the depth of the AW core does 593 not always reflect the depth of the top of the AW layer, particularly in the eastern Arctic, and care must be taken when using temporal trends in AW core properties to assess 595 trends in the AW as a whole - a cooling or increase in salinity of the core does not nec-596 essarily translate to a cooling or increase in salinity of the entire AW layer. 597

In general, as the AW is advected around the Arctic the potential temperature and 598 salinity of its core decrease. Despite freshening, the AW core density increases along its 599 advection pathway. This is partially due to the preferential loss of heat and salt from 600 the top of the AW layer to the fresher, cooler water above through vertical mixing along 601 the AW advection pathway. This likely deepens the core without the AW layer as a whole 602 getting denser - upper AW cools such that the AW core (temperature maximum) is found 603 on deeper (denser) isopycnals. Interaction with dense shelf flows formed by brine rejec-604 tion during sea-ice formation may also play an important role in the cooling and fresh-605 ening of the AW core during its advection around the basin. 606

The evolution of AW has differed between the eastern and western basins of the 607 Arctic. In the Eurasian Basin, AW core temperature and AW heat content increased from 608 2002-2018, with the former increasing by approximately 0.7 °C during this period. Warm 609 pulses were superimposed upon this trend. Instances of high upper ocean heat content 610 in the east were found to be associated with shallower, fresher AW. In contrast to this, 611 and similar reports in the literature of eastern Arctic Atlantification, the western Arc-612 tic saw AW core temperatures decrease from a previous warm peak by approximately 613 $0.1 \ ^{\circ}C$ between 2008–2018 (although AW layer heat content increased), and also saw AW 614 heat become more isolated from the surface. This increased isolation was due to Beau-615 fort Gyre intensification which deepened the halocline. These findings suggest the emer-616 gence of two different regimes - with AW affecting sea-ice in the east, and Pacific Wa-617 ter influencing sea-ice in the west. This implies that the future evolution of the Eurasian 618 Basin will strongly depend on AW, whereas Pacific Water and the Beaufort Gyre will 619 be the biggest drivers of change in the Canada Basin. This contrasting regional evolu-620 tion is in agreement with other recent studies, which describe halocline weakening, AW 621 shoaling, and increased sub-Arctic influence in the Eurasian Basin, contrasting with a 622

freshening and deepening of the surface layer in the Amerasian Basin driven by local atmospheric conditions (Polyakov, Rippeth, et al., 2020; Polyakov, Alkire, et al., 2020).

Despite the limitation of sparse, temporally inhomogeneous oceanographic mea-625 surements in the Arctic, pan-Arctic observational analysis can give useful insights into 626 the overall temporal and spatial patterns of heat distribution in the Arctic Ocean. Given 627 the challenges of realistically representing the AW layer in forced ocean-sea-ice and cou-628 pled climate models, and the stark regional differences emerging in the Arctic Ocean, 629 the use of pan-Arctic observations for model validation and benchmarking will be essen-630 631 tial. Only by combining insight from observations and models will we be able to accurately determine what the future Arctic will look like under a changing climate, which 632 is important both for the region itself as well as for the wider climate system. 633

⁶³⁴ Open Research

Profile data used in this study are available from http://www.whoi.edu/itp (for ITP data), https://www.whoi.edu/beaufortgyre (for BGEP data), https://uaf-iarc .org/nabos (for NABOS data), and https://www.ncei.noaa.gov/products/world-ocean -database (for WOD data). The Atlantic Water core data computed for this study are available as an Open Access dataset at the Oxford University Research Archive via https:// doi.org/10.5287/bodleian:wxv8GA7Mk. Note that this dataset includes all mooring data rather than monthly mean mooring data.

642 Acknowledgments

This study was funded by the Natural Environment Research Council (NERC) via a DTP studentship under grant number NE/L002612/1. HLJ is grateful for funding from the NERC ArctiCONNECT project (grant number NE/V005855/1). We thank the two anonymous reviewers and associate editor for their useful insights and constructive comments.

647 **References**

| 648 | Aksenov, Y., Ivanov, V. V., Nurser, A. J. G., Bacon, S., Polyakov, I. V., Coward, |
|-----|--|
| 649 | A. U., Beszczynska-Moeller, A. (2011). The Arctic Orcumpolar Boundary C_{1} |
| 650 | Current. J. Geophys. Res. Ocean., 116(9). doi: 10.1029/2010JC006637 |
| 651 | Anderson, L. G., Bjrrk, G., Jones, E. P., Kattner, G., Koltermann, K. P., Liljeblad, |
| 652 | B., Swift, J. (1994). Results from the Oden 91 expedition. , 99 , 3273 – |
| 653 | 3283. |
| 654 | Barton, B. I., Lenn, Y. D., & Lique, C. (2018). Observed Atlantification of the Bar- |
| 655 | ents Sea Causes the Polar Front to Limit the Expansion of Winter Sea Ice. J. |
| 656 | Phys. Oceanogr., 48(8), 1849–1866. doi: 10.1175/JPO-D-18-0003.1 |
| 657 | Bebieva, Y., & Timmermans, M. L. (2017). The relationship between double- |
| 659 | diffusive intrusions and staircases in the Arctic Ocean I Phys Oceanoor |
| 650 | $\sqrt{2}(4)$ 867-878 doi: 10.1175/JPO-D-16-0265.1 |
| 059 | Pabiers V & Timmermang M I (2010) Double Diffusive Largering in the Canada |
| 660 | Debieva, I., & Timmermans, M. L. (2019). Double-Diffusive Layering in the Canada |
| 661 | Basin: An Explanation of Along-Layer Temperature and Salinity Gradients. J. |
| 662 | Geophys. Res. Ocean., 124(1), 723–735. doi: 10.1029/2018JC014368 |
| 663 | Beszczynska-Moller, A., Fahrbach, E., Schauer, U., & Hansen, E. (2012). Variabil- |
| 664 | ity in Atlantic water temperature and transport at the entrance to the Arctic |
| 665 | Ocean, 1997–2010. J. Mar. Sci., 69(5), 852–863. doi: 10.1038/278097a0 |
| 666 | Carmack, E., Polyakov, I., Padman, L., Fer, I., Hunke, E., Hutchings, J., Win- |
| 667 | sor, P. (2015). Toward quantifying the increasing role of oceanic heat in sea |
| 668 | ice loss in the new Arctic. Bull. Am. Meteorol. Soc., 96(12), 2079–2105. doi: |
| 669 | 10.1175/BAMS-D-13-00177.1 |
| 670 | Dmitrenko, I. A., Polyakov, I. V., Kirillov, S. A., Timokhov, L. A., Frolov, I. E., |

| 671 | Sokolov, V. T., Walsh, D. (2008). Toward a warmer Arctic Ocean: |
|-----|---|
| 672 | Spreading of the early 21st century Atlantic water warm anomaly along |
| 673 | the Eurasian basin margins. J. Geophys. Res. Ocean., 113(5), 1–13. doi: |
| 674 | 10.1029/2007JC004158 |
| 675 | Fer, I. (2009). Weak Vertical Diffusion Allows Maintenance of Cold Halocline in the |
| 676 | Central Arctic. Atmos. Ocean. Sci. Lett., 2(3), 148–152. doi: 10.1080/16742834 |
| 677 | .2009.11446789 |
| 678 | Ilıcak, M., Drange, H., Wang, Q., Gerdes, R., Aksenov, Y., Bailey, D., Yeager, |
| 679 | S. G. (2016, apr). An assessment of the Arctic Ocean in a suite of interannual |
| 680 | CORE-II simulations. Part III: Hydrography and fluxes. Ocean Model., 100, |
| 681 | 141–161. doi: 10.1016/j.ocemod.2016.02.004 |
| 682 | IOC, SCOR, & IAPSO. (2010). The International Thermodynamic Equation of |
| 683 | Seawater-2010: Calculation and Use of Inermodynamic Properties, Manuals |
| 684 | Guilles. Paris: UNESCO. |
| 685 | Tracing Atlantic Water Signature in the Arctic Sea Ice Cover East of Svalhard |
| 687 | Adv. Meteorol 2012. doi: 10.1155/2012/201818 |
| 688 | Ivanov V V & Golovin P N (2007) Observations and modeling of dense water |
| 689 | cascading from the northwestern Laptev Sea shelf. J. Geophys. Res. Ocean. |
| 690 | 112(9), 1–15. doi: 10.1029/2006JC003882 |
| 691 | Ivanov, V. V., Shapiro, G. I., Huthnance, J. M., Aleynik, D. L., & Golovin, P. N. |
| 692 | (2004). Cascades of dense water around the world ocean (Vol. 60) (No. 1). doi: |
| 693 | 10.1016/j.pocean.2003.12.002 |
| 694 | Jackson, J. M., Williams, W. J., & Carmack, E. C. (2012). Winter sea-ice melt in |
| 695 | the Canada Basin, Arctic Ocean. Geophys. Res. Lett., $39(3)$, 2–7. doi: 10 |
| 696 | .1029/2011 GL 050219 |
| 697 | Karcher, M., Beszczynska-Möller, A., Kauker, F., Gerdes, R., Heyen, S., Rudels, |
| 698 | B., & Schauer, U. (2011). Arctic Ocean warming and its consequences for |
| 699 | the Denmark Strait overflow. J. Geophys. Res. Ocean., $11b(2)$, 1–10. doi: 10.1020/2010. C006265 |
| 700 | Karcher M. Smith I. N. Kauker F. Cordes R. & Smethie W. M. (2012). Recent |
| 701 | changes in Arctic Ocean circulation revealed by iodine-129 observations and |
| 702 | modeling. J. Geophys. Res. Ocean., 117(8), 1–17. doi: 10.1029/2011JC007513 |
| 704 | Karcher, M. J., Gerdes, R., Kauker, F., & Köberle, C. (2003). Arctic warm- |
| 705 | ing: Evolution and spreading of the 1990s warm event in the Nordic seas |
| 706 | and the Arctic Ocean. J. Geophys. Res. C Ocean., 108(2), 16–1. doi: |
| 707 | 10.1029/2001jc001265 |
| 708 | Karcher, M. J., & Oberhuber, J. (2002). Pathways and modification of the upper |
| 709 | and intermediate waters of the Arctic Ocean. J. Geophys. Res., 107(C6), 1–13. |
| 710 | doi: 10.1029/2000jc000530 |
| 711 | Krishfield, R., Toole, J., Proshutinsky, A., & Timmermans, M. L. (2008). Au- |
| 712 | tomated Ice-Tethered Profilers for Seawater Observations under Pack Ice |
| 713 | in All Seasons. J. Atmos. Ocean. Technol., 25(11), 2091–2105. doi: 10.1175/2000/JEDCHO507.1 |
| 714 | $\frac{10.1175}{2008J1ECHO387.1}$ |
| 715 | dynamical features of intrusive layering in the Eurosian Basin in the Aratia |
| 716 | Ocean I Geonbus Res Ocean 116(12) 1–15 doi: 10.1029/2010.IC006920 |
| 718 | Ladd, C., Mordy, C. W., Salo, S. A. & Stabeno, P. J. (2016) Winter Water Pron- |
| 719 | erties and the Chukchi Polynva. J. Geonhus. Res. Ocean. 121(8), 5516–5534 |
| 720 | doi: 10.1038/175238c0 |
| 721 | Li, J., Pickart, R. S., Lin, P., Bahr, F., Arrigo, K. R., Juranek, L., & Yang, X. Y. |
| 722 | (2020). The Atlantic Water Boundary Current in the Chukchi Borderland |
| 723 | and Southern Canada Basin. J. Geophys. Res. Ocean., 125(8), 1–20. doi: |
| 724 | $10.1029/2020 \mathrm{JC016197}$ |
| 725 | Lind, S., Ingvaldsen, R. B., & Furevik, T. (2018, jul). Arctic warming hotspot in the |

| 726 | northern Barents Sea linked to declining sea-ice import. Nat. Clim. Chang., |
|------------|--|
| 727 | 8(7), 634-639. doi: 10.1038/s41558-018-0205-y |
| 728 | Lique, C., & Johnson, H. L. (2015, nov). Is there any imprint of the wind variabil- |
| 729 730 | Ity on the Atlantic Water circulation within the Arctic Basin? Geophys. Res. Lett., $42(22)$, 9880–9888. doi: 10.1002/2015GL066141 |
| 731 | Lique, C., Johnson, H. L., & Davis, P. E. D. (2015, may). On the Interplay be- |
| 732 | tween the Circulation in the Surface and the Intermediate Layers of the Arctic |
| 733 | Ocean. J. Phys. Oceanogr., 45(5), 1393–1409. doi: 10.1175/JPO-D-14-0183.1 |
| 734 | Lique, C., Johnson, H. L., & Plancherel, Y. (2018). Emergence of deep convection in |
| 735 | the Arctic Ocean under a warming climate. Clim. Dyn., 50(9-10), 3833–3847. |
| 736 | doi: 10.1007/s00382-017-3849-9 |
| 737 | Lique, C., & Steele, M. (2012, mar). Where can we find a seasonal cycle of the |
| 738 | Atlantic water temperature within the Arctic Basin? J. Geophys. Res. Ocean., |
| 739 | 117(C3), n/a–n/a. doi: 10.1029/2011JC007612 |
| 740 | Lique, C., Treguier, A. M., Blanke, B., & Grima, N. (2010). On the origins of water |
| 741 | masses exported along both sides of Greenland: A Lagrangian model analysis. |
| 742 | J. Geophys. Res. Ocean., 115(5), 1–20. doi: 10.1029/2009JC005510 |
| 743 | Luneva, M. V., Ivanov, V. V., Iuzov, F., Aksenov, T., Harle, J. D., Keny, S., & Holt, J. T. (2020) Hotapota of Dance Water Caseding in the Aretic |
| 744 | Quantity In the Arctic Case of the Parific Water Dathways I Compare Real Organ |
| 745 | 195(10) doi: 10.1020/2020 IC016044 |
| 740 | McLaughlin F A Carmack F C Williams W I Zimmormann S Shimada |
| 747 | K k Itoh M (2000) Loint effects of boundary currents and thermobaline |
| 748 | intrusions on the warming of Atlantic water in the Canada Basin 1993-2007 |
| 750 | J Geophys Res Ocean 11/(7) doi: 10.1029/2008JC005001 |
| 751 | Melling H & Moore B (1995) Modification of halocline source waters during |
| 752 | freezing on the Beaufort Sea shelf: evidence from oxygen isotopes and dis- |
| 753 | solved nutrients. Science (80-,)., 15(1), 89–113. |
| 754 | Pérez-Hernández, M. D., Pickart, R. S., Torres, D. J., Bahr, F., Sundfiord, A., In- |
| 755 | gvaldsen, R., Pavlov, V. (2019). Structure, Transport, and Seasonality |
| 756 | of the Atlantic Water Boundary Current North of Svalbard: Results From a |
| 757 | Yearlong Mooring Array. J. Geophys. Res. Ocean., 124(3), 1679–1698. doi: |
| 758 | 10.1029/2018JC014759 |
| 759 | Perovich, D. K., & Richter-Menge, J. A. (2015). Regional variability in sea ice |
| 760 | melt in a changing Arctic. Philos. Trans. R. Soc. A Math. Phys. Eng. Sci., |
| 761 | 373(2045). doi: 10.1098/rsta.2014.0165 |
| 762 | Polyakov, I. V., Alekseev, G. V., Timokhov, L. A., Bhatt, U. S., Colony, R. L., Sim- |
| 763 | mons, H. L., Zakharov, V. F. (2004). Variability of the Intermediate |
| 764 | Atlantic Water of the Arctic Ocean over the Last 100 Years. J. Clim., $17(23)$, |
| 765 | 4485–4497. doi: 10.1175/JCLI-3224.1 |
| 766 | Polyakov, I. V., Alexeev, V. A., Ashik, I. M., Bacon, S., Beszczynska-Möller, |
| 767 | A., Carmack, E. C., Woodgate, R. (2011). Fate of Early 2000s Arc- |
| 768 | tic Warm Water Pulse. Bull. Am. Meteorol. Soc., 92(5), 561–566. doi: |
| 769 | 10.1175/2010BAMS2921.1 |
| 770 | Polyakov, I. V., Alkire, M. B., Bluhm, B. A., Brown, K. A., Carmack, E. C., |
| 771 | Chierici, M., Wassmann, P. (2020). Borealization of the Arctic Ocean |
| 772 | in Response to Anomalous Advection From Sub-Arctic Seas. Front. Mar. Sci., |
| 773 | 7(July). doi: 10.3389/tmars.2020.00491 |
| 774 | Polyakov, I. V., Beszczynska, A., Carmack, E. C., Dmitrenko, I. A., Fahrbach, E., |
| 775 | Frolov, I. E., Walsh, J. E. (2005). One more step toward a warmer Arctic. |
| 776 | Geophys. Kes. Lett., 32(17), 1–4. doi: 10.1029/2005GL023740 |
| 777 | Polyakov, I. V., Pnyushkov, A. V., Alkire, M. B., Ashik, I. M., Baumann, T. M., |
| 778 | Carmack, E. C., runn, A. (2017, apr). Greater role for Atlantic inflows |
| 779 | on sea-ice loss in the Eurasian Basin of the Arctic Ocean. Science (80)., 256(6225) 285 201 doi: 10.1126/acicros.cc. 22004 |
| 780 | 350(0355), 205-231, 001, 10.1120/science.aato204 |

| 781 | Polyakov, I. V., Pnyushkov, A. V., & Carmack, E. C. (2018). Stability of the arc- |
|---|---|
| 782 | tic halocline: A new indicator of arctic climate change. Environ. Res. Lett., |
| 783 | 13(12), 125008. doi: $10.1088/1748-9326/aaec1e$ |
| 784 | Polyakov, I. V., Pnyushkov, A. V., & Timokhov, L. A. (2012, dec). Warming of |
| 785 | the Intermediate Atlantic Water of the Arctic Ocean in the 2000s. J. Clim., |
| 786 | 25(23), 8362-8370. doi: 10.1175/JCLI-D-12-00266.1 |
| 787 | Polyakov, I. V., Rippeth, T. P., Fer, I., Alkire, M. B., Baumann, T. M., Carmack, |
| 788 | E. C., Rember, R. (2020). Weakening of Cold Halocline Layer Exposes Sea |
| 789 | Ice to Oceanic Heat in the Eastern Arctic Ocean. J. Clim., 33(18), 8107–8123. |
| 790 | doi: 10.1175/JCLI-D-19-0976.1 |
| 791 | Polyakov, I. V., Timokhov, L. A., Alexeev, V. A., Bacon, S., Dmitrenko, I. A., |
| 792 | Fortier, L., Toole, J. (2010). Arctic Ocean Warming Contributes |
| 793 | to Reduced Polar Ice Cap. J. Phys. Oceanogr., $40(12)$, 2743–2756. doi: |
| 794 | 10.1175/2010JPO4339.1 |
| 795 | Proshutinsky, A., Krishfield, R., Toole, J. M., Timmermans, M. L., Williams, W., |
| 796 | Zimmermann, S., Zhao, J. (2019). Analysis of the Beaufort Gyre Freshwa- |
| 797 | ter Content in 2003–2018. J. Geophys. Res. Ocean., $124(12)$, 9658–9689. doi: |
| 798 | 10.1029/2019JC015281 |
| 799 | Regan, H. C., Lique, C., & Armitage, T. W. (2019). The Beaufort Gyre Extent, |
| 800 | Shape, and Location Between 2003 and 2014 From Satellite Observations. J. |
| 801 | Geophys. Res. Ocean., 124(2), 844–862. doi: 10.1029/2018JC014379 |
| 802 | Ruddick, B. (1992). Intrusive Mixing in a Mediterranean Salt Lens - Intru- |
| 803 | sion Slopes and Dynamical Mechanisms (Vol. 22) (No. 11). doi: 10.1175/ |
| 804 | 1520-0485(1992)022(1274:1MIAMS)2.0.CO;2 |
| 805 | Rudels, B. (2015). Arctic Ocean circulation, processes and water masses: A de- |
| 806 | scription of observations and ideas with focus on the period prior to the L_{1} |
| 807 | International Polar Year 2007-2009. Prog. Oceanogr., 132, 22–67. doi: |
| | 10, 1016 /; masses $9012, 11, 006$ |
| 808 | 10.1016/j.pocean.2013.11.006 |
| 808 809 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Perents and Kam Seeg. Deep Res. Bart I. Oceanogr |
| 808 809 810 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. <i>Deep. Res. Part I Oceanogr.</i> <i>Res. Part J Oceanogr.</i> 10.1016/S0067_0627(02)00125_5 |
| 808 809 810 811 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, O., Wang, O., Su, L. Li, X., & Ciao, F. (2010). Association of the Atlantia way. |
| 808 809 810 811 812 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dava, 52(9, 10) |
| 808 809 810 811 812 813 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. <i>Deep. Res. Part I Oceanogr. Res. Pap.</i>, 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. <i>Clim. Dyn.</i>, 53(9-10), 5270–5291. doi: 10.1007/s00382.019-04870.6 |
| 808 809 810 811 812 813 814 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skageth Ø. Elderik, T. Årthun M. Ashigrneon H. Lion V. S. & Smoderud |
| 808 809 810 811 812 813 814 815 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Roduced officiency of the Barents Sea cooling machine. Nat. |
| 808 809 810 811 812 813 814 815 816 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. <i>Deep. Res. Part I Oceanogr. Res. Pap.</i>, 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. <i>Clim. Dyn.</i>, 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. <i>Nat. Clim. Chang.</i> 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 |
| 808 809 810 811 812 813 814 815 816 817 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielbagen B. F. Werner K. Sørensen S. A. Kandiano, F. Budeus G. Husum |
| 808 809 810 811 812 813 814 815 816 817 818 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K. & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80-1), 331(6016), 450–453. |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield B |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., M., Nishino, S. (2014). Mechanisms of Pacific Summer Water |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean. |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 826 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 825 826 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53 (9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 824 825 826 827 828 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 826 827 828 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. doi: 10.1126/sciadv.aat6773 |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 826 827 828 829 830 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. doi: 10.1126/sciadv.aat6773 Toole, J. M., Krishfield, R. A., Timmermans, M. L., & Proshutinsky, A. (2011). The |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 822 823 824 825 826 827 828 829 830 831 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. doi: 10.1126/sciadv.aat6773 Toole, J. M., Krishfield, R. A., Timmermans, M. L., & Proshutinsky, A. (2011). The Ice-Tethered profiler: Argo of the Arctic. Oceanography, 24(3), 126–135. doi: |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 825 826 827 828 829 830 831 832 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331 (6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. doi: 10.1126/sciadv.aat6773 Toole, J. M., Krishfield, R. A., Timmermans, M. L., & Proshutinsky, A. (2011). The Ice-Tethered profiler: Argo of the Arctic. Oceanography, 24(3), 126–135. doi: 10.5670/oceanog.2011.64 |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 826 827 828 829 830 831 832 833 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523-7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. doi: 10.1126/sciadv.aat6773 Toole, J. M., Krishfield, R. A., Timmermans, M. L., & Proshutinsky, A. (2011). The Ice-Tethered profiler: Argo of the Arctic. Oceanography, 24(3), 126–135. doi: 10.5670/oceanog.2011.64 Turner, J. S. (2010, jan). The Melting of Ice in the Arctic Ocean: The Influence |
| 808 809 810 811 812 813 814 815 816 817 818 819 820 821 822 823 824 825 826 827 828 829 830 831 832 833 834 | 10.1016/j.pocean.2013.11.006 Schauer, U., Loeng, H., Rudels, B., Ozhigin, V. K., & Dieck, W. (2002). Atlantic Water flow through the Barents and Kara Seas. Deep. Res. Part I Oceanogr. Res. Pap., 49(12), 2281–2298. doi: 10.1016/S0967-0637(02)00125-5 Shu, Q., Wang, Q., Su, J., Li, X., & Qiao, F. (2019). Assessment of the Atlantic water layer in the Arctic Ocean in CMIP5 climate models. Clim. Dyn., 53(9-10), 5279–5291. doi: 10.1007/s00382-019-04870-6 Skagseth, Ø., Eldevik, T., Årthun, M., Asbjørnsen, H., Lien, V. S., & Smedsrud, L. H. (2020). Reduced efficiency of the Barents Sea cooling machine. Nat. Clim. Chang., 10(7), 661–666. doi: 10.1038/s41558-020-0772-6 Spielhagen, R. F., Werner, K., Sørensen, S. A., Kandiano, E., Budeus, G., Husum, K., & Marchitto, T. M. (2011). Enhanced Modern Heat Transfer to the Arctic by Warm Atlantic Water. Science (80)., 331(6016), 450–453. Timmermans, M. L. (2015). The impact of stored solar heat on Arctic sea ice growth. Geophys. Res. Lett., 42(15), 6399–6406. doi: 10.1002/2015GL064541 Timmermans, M. L., Proshutinsky, A., Golubeva, E., Jackson, J. M., Krishfield, R., McCall, M., Nishino, S. (2014). Mechanisms of Pacific Summer Water variability in the Arctic's Central Canada Basin. J. Geophys. Res. Ocean., 119, 7523–7548. doi: 10.1038/175238c0 Timmermans, M. L., Toole, J., & Krishfield, R. (2018). Warming of the interior Arctic Ocean linked to sea ice losses at the basin margins. Sci. Adv., 4(8), 1–7. doi: 10.1126/sciadv.aat6773 Toole, J. M., Krishfield, R. A., Timmermans, M. L., & Proshutinsky, A. (2011). The Ice-Tethered profiler: Argo of the Arctic. Oceanography, 24(3), 126–135. doi: 10.5670/oceanog.2011.64 Turner, J. S. (2010, jan). The Melting of Ice in the Arctic Ocean: The Influence of Double-Diffusive Transport of Heat from Below. J. Phys. Oceanogr., 40(1), |

| 836 | Wefing, A. M., Casacuberta, N., Christl, M., Gruber, N., & Smith, J. N. (2020). |
|-----|---|
| 837 | Circulation timescales of Atlantic Water in the Arctic Ocean determined |
| 838 | from anthropogenic radionuclides. $Ocean Sci., 17(1), 111-129.$ doi: |
| 839 | 10.5194 / os- 17 - 111 - 2021 |
| 840 | Woodgate, R. A., Aagaard, K., Muench, R. D., Gunn, J., Björk, G., Rudels, B., |
| 841 | Schauer, U. (2001). The Arctic Ocean Boundary Current along the Eurasian |
| 842 | slope and the adjacent Lomonosov Ridge: Water mass properties, transports |
| 843 | and transformations from moored instruments. Deep. Res. Part I Oceanogr. |
| 844 | Res. Pap., $48(8)$, 1757–1792. doi: 10.1016/S0967-0637(00)00091-1 |
| 845 | Zhang, J., Steele, M., Runciman, K., Dewey, S., Morison, J., Lee, C., Toole, J. |
| 846 | (2016, nov). The Beaufort Gyre intensification and stabilization: A model- |
| 847 | observation synthesis. J. Geophys. Res. Ocean., 121(11), 7933–7952. doi: |
| 848 | 10.1002/2016JC012196 |
| | Zhong W & Zhoo I (2014) Depending of the Atlantic Water Core in the Canada |

Zhong, W., & Zhao, J. (2014). Deepening of the Atlantic Water Core in the Canada
 Basin in 2003-11. J. Phys. Oceanogr., 44 (9), 2353–2369. doi: 10.1175/JPO-D
 -13-084.1